Author's response to reviews

Title: A proteomic strategy to identify novel serum biomarkers for liver cirrhosis and hepatocellular cancer in individuals with fatty liver disease.

Authors:

Joe Gray (joe.gray@ncl.ac.uk)
Dipankar Chattopadhyay (dchatto@doctors.org.uk)
Beale S Gary (gary.beale@ncl.ac.uk)
Gillian L Patman (Gillian.Patman@ncl.ac.uk)
Luca Miele (luca.miele@rm.unicatt.it)
Barry P King (B.P.King@ncl.ac.uk)
Stewart Stephen (Stephen.Stewart@nuth.nhs.uk)
Hudson Mark (Mark.Hudson@nuth.nhs.uk)
Christopher P Day (c.p.day@ncl.ac.uk)
Derek M Manas (Derek.Manas@nuth.nhs.uk)
Helen L Reeves (H.L.Reeves@ncl.ac.uk)

Version: 3 Date: 4 August 2009

Author's response to reviews:

Thankyou for accepting our manuscript for publication. Both major and minor corrections, as detailed by email on 03 August 2009 (pasted below for your reference) have been made to the revised manuscript which has been uploaded today.

Helen Reeves

Formatting changes requested
-----------------------------

Major revisions

Author list - The list of authors in the manuscript has been re-written exactly as in the submission system.

Affiliations - Each affiliation has been written in full.

Reference 8 has been corrected.

Reference 18 has been corrected.

Additional files - The section titled "Additional files" at the end of the manuscript has been ammended as advised for each of the 11 files.

Additional files - have been mentioned in the text.

Minor revisions
Title - Has been revised using sentence case
Table title and legend - The table titles and legends have been revised.
Tables - vertical lines have been removed.
The manuscript has been checked and a small number of typographical errors have been amended.